

Serological Evidence of Contrasted Exposure to Arboviral Infections between Islands of the Union of Comoros (Indian Ocean)

Koussay Dellagi, Nicolas Salez, Marianne Maquart, Sophie Larrieu, Amina Yssouf, Rahamatou Silaï, Isabelle Leparc-Goffart, Pablo Tortosa, Xavier de Lamballerie

▶ To cite this version:

Koussay Dellagi, Nicolas Salez, Marianne Maquart, Sophie Larrieu, Amina Yssouf, et al.. Serological Evidence of Contrasted Exposure to Arboviral Infections between Islands of the Union of Comoros (Indian Ocean). PLoS Neglected Tropical Diseases, 2016, 10 (12), pp.e0004840. 10.1371/journal.pntd.0004840. hal-01423540

HAL Id: hal-01423540 https://hal.science/hal-01423540v1

Submitted on 5 Oct 2018 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



Citation: Dellagi K, Salez N, Maquart M, Larrieu S, Yssouf A, Silaï R, et al. (2016) Serological Evidence of Contrasted Exposure to Arboviral Infections between Islands of the Union of Comoros (Indian Ocean). PLoS Negl Trop Dis 10(12): e0004840. doi:10.1371/journal.pntd.0004840

Editor: Patricia V. Aguilar, University of Texas Medical Branch, UNITED STATES

Received: May 21, 2015

Accepted: June 20, 2016

Published: December 15, 2016

Copyright: © 2016 Dellagi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This study was supported by institutional funds provided to CRVOI by the French Ministry of High Education and Research. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

Serological Evidence of Contrasted Exposure to Arboviral Infections between Islands of the Union of Comoros (Indian Ocean)

Koussay Dellagi^{1,2}*, Nicolas Salez³, Marianne Maquart⁴, Sophie Larrieu⁵, Amina Yssouf⁶, Rahamatou Silaï⁶, Isabelle Leparc-Goffart⁴, Pablo Tortosa^{1,2}, Xavier de Lamballerie³

1 Centre de Recherche et de Veille sur les Maladies Emergentes dans l'Océan Indien, (CRVOI) Plateforme de Recherche CYROI, Sainte Clotilde, Reunion Island, France, 2 Unité Mixte de Recherche «Processus Infectieux en Milieu Insulaire Tropical» (UMR PIMIT), INSERM 1187 CNRS 9192 IRD 249 Université de La Réunion, Plateforme de Recherche CYROI, Sainte Clotilde, La Réunion, France, 3 Aix Marseille Université, IRD, EHESP French School of Public Health, EPV UMR_D 190 "Emergence des Pathologies Virales", Marseille, France, 4 French National Reference Centre for Arbovirus, IRBA, Marseille, France, 5 Cellule Interrégionale d'Épidémiologie Océan Indien (Cire OI), Institut de Veille Sanitaire, Saint Denis, La Réunion, France, 6 Programme National de Lutte contre le Paludisme, Moroni, Union of the Comoros

* koussay.dellagi@ird.fr

Abstract

A cross sectional serological survey of arboviral infections in humans was conducted on the three islands of the Union of Comoros, Indian Ocean, in order to test a previously suggested contrasted exposure of the three neighboring islands to arthropod-borne epidemics. Four hundred human sera were collected on Ngazidja (Grande Comore), Mwali (Mohéli) and Ndzouani (Anjouan), and were tested by ELISA for IgM and/or IgG antibodies to Dengue (DENV), Chikungunya (CHIKV), Rift Valley fever (RVFV), West Nile (WNV), Tick borne encephalitis (TBEV) and Yellow fever (YFV) viruses and for neutralizing antibodies to DENV serotypes 1-4. Very few sera were positive for IgM antibodies to the tested viruses indicating that the sero-survey was performed during an inter epidemic phase for the investigated arbovirus infections, except for RVF which showed evidence of recent infections on all three islands. IgG reactivity with at least one arbovirus was observed in almost 85% of tested sera, with seropositivity rates increasing with age, indicative of an intense and long lasting exposure of the Comorian population to arboviral risk. Interestingly, the positivity rates for IgG antibodies to DENV and CHIKV were significantly higher on Ngazidja, confirming the previously suggested prominent exposure of this island to these arboviruses, while serological traces of WNV infection were detected most frequently on Mwali suggesting some transmission specificities associated with this island only. The study provides the first evidence for circulation of RVFV in human populations from the Union of Comoros and further suggests that the virus is currently circulating on the three islands in an inconspicuous manner. This study supports contrasted exposure of the islands of the Comoros archipelago to arboviral infections. The observation is discussed in terms of ecological factors that may affect the abundance and distribution of vector populations on the three islands as well as concurring anthropogenic factors that may impact arbovirus transmission in this diverse island ecosystem.

Author Summary

Peculiarities of Island ecosystems make oceanic islands most suitable areas for analyzing how variations in local ecological conditions may impact on the intensity of arboviral transmission. The Comoros archipelago (2144 Km2) consists of four volcanic islands that are distant of 40-60 Km apart, located at the northern end of the Mozambique Channel. The three northern islands constitute the Union of Comoros that ranks amongst the least developed countries in the world. The geographic proximity to Africa exposes the Comoros archipelago to arboviral emergence as evidenced by a number of epidemics recorded in the last decades (i.e. dengue, chikungunya, and Rift Valley fever). Most interestingly, there is historical evidence that these epidemics have had contrasted courses and severity on the 3 islands of the Union of the Comoros, despite their cultural homogeneity and the frequent inter island mobility of people and livestock. Based on these observations, we conducted a serological investigation in order to delineate the risk to arboviral infection on the 3 islands of the Union of Comoros and assess whether it is relevant to all arboviral species or only to some of them. Our results confirm the differential exposure of the Comoros islands to the arboviral risk. The ecological and anthropological factors that may account for this contrasted epidemiology are discussed.

Introduction

Vector-borne infections are mostly sensitive to environmental changes whether natural or anthropogenic. Slight variations in ecologic conditions may severely affect pathogen transmission capacity by impacting the diversity, abundance and/or behavior of vectors, their competence for transmission, together with several traits of the parasite itself [1,2]. Unraveling these factors helps understanding how vector borne diseases may express contrasted dynamics in different geographic locations and identifying drivers of emergence acting either at local, regional or large distance scales. Island ecosystems provide ideal *in natura* conditions allowing to uncouple local transmission from long distance spread [3]. The peculiarities of such ecosystems (geographic isolation, limited land surface area, low species richness and often-high levels of endemism), make oceanic islands most suitable for comprehensive entomological surveys and investigations of original host/pathogen interactions [3].

The Comoros archipelago (2144 Km2) consists of four volcanic islands that have emerged *de novo* from the South-Western Indian Ocean (SWIO) floor at the Comoros hot spot. They are distant of 40–60 Km from each other and are aligned on a Northwest to Southeast line, at the north end of the Mozambique Channel. The three northern islands constitute the Union of Comoros that ranks amongst the least developed countries in the world. The southern island of Mayotte is administered by France and exhibits significantly higher development indices.

The islands of the Union of Comoros are small and sparsely populated. The largest island (1146Km2) sheltering the capital is named Ngazidja (Grande Comore in French) and has 380 000 inhabitants. The second larger island, Ndzouani (Anjouan) (424 Km2) has 300 000 inhabitants, while Mwali (Moheli) is the smallest island (290Km2) and is home to less than 45 000 inhabitants.

The geographic proximity to Africa exposes the Comoros archipelago to arboviral emergence as evidenced by a number of epidemics recorded in the last decades. Hence, dengue, chikungunya and Rift Valley fever outbreaks flared up on the Comoros islands in first place before spreading to other islands of the SWIO. Most interestingly, there is evidence that these epidemics have had contrasted courses and severity on the 3 islands of the Union of the Comoros, despite their cultural homogeneity and the frequent inter island mobility of people and livestock.

The different exposure of the three islands to the arboviral risk is most evidenced by the last epidemics of dengue and chikungunya. In 1993, a large epidemic due to Dengue virus serotype 1 (DENV1) affected Ngazidja, with a prevalence of 26% of IgM antibodies to Dengue virus (DENV) [4]. In contrast, the outbreak was much less severe on the sister islands of Ndzouani and Mwali where a ten times lower IgM positivity rate was detected [4], and no dengue epidemic was reported at the same time on Mayotte. In March 2010, a Dengue virus serotype 3 (DENV3) epidemic was reported on Ngazidja where 1805 suspected cases were registered while the number of suspected cases was only 18 and 4 on Mwali and Ndzouani respectively, as informally reported by the Comorian Health Authorities (www.reseausega-coi.org/system/files/06-Mlindasse-DENV3_0.pdf15). Only 76 cases of dengue were confirmed on Mayotte despite an efficient surveillance system [5].

In 2004, a CHIK outbreak flared up in Lamu, Kenya [6] then spread to Ngazidja where 5202 cases were reported from January to May 2005. A sero-survey conducted on the island during the epidemic showed an attack rate of 63% with 60% IgM and 27% IgG positive samples [7] According to the Comorian Health Authorities [8], the outbreak appeared much less severe on the other Comorian islands with 207 cases reported on Ndzouani and only 1 on Mwali. On Mayotte, the epidemic curve delineated two waves [9]: the first wave (April-June 2005) was concomitant to the epidemic on Ngazidja but of minor intensity as assessed by the very low prevalence (1.6%) detected on October 2005; the second wave was however explosive: it started on early 2006 and peaked on March-April with a dramatic rise of CHIK virus (CHIKV) sero-conversion rate. This second wave then expanded into a huge epidemic that involved all the SWIO islands, *i.e.* Madagascar, Reunion, Mauritius, and Seychelles [10]. Interestingly the second explosive outbreak completely spared the three islands of Union of Comoros where the CHIK dynamics was unimodal.

Though previous investigations suggest a differentiated impact of arboviral diseases on the three islands, the available information supporting this view are of unequal solidity with regard to the three islands. Hence, in order to get an unequivocal assessment of this point, we realized a serological investigation in the Union of Comoros in order to delineate the arboviral risk and check whether this assumption is relevant to all viral species or only to some of them.

Materials and Methods

Ethics statement

The present study was implemented at the initiative of Health authorities of the Union of Comoros (Ministère de la Santé, de la Solidarité et de la Promotion du Genre) in order to evaluate the impact of vector borne diseases on the general population. The National Malaria Control Program (PNLP) was identified by the Health authorities as the promoter of the study to supervise the serological investigation according to a protocol ensuring that all samples were anonymized and the study conducted ethically (#1175/MSSPG/DNS). The present study used excess sera collected from the PNLP and from private laboratories established on Ngazidja, Ndzouani and Mwali. The PNLP has approved that only an oral informed consent should be required from the participants, in agreement with the Comoros cultural norms. The consent was collected by one of the co-authors (RS), who is member of the PNLP and explained the objectives of the investigation, how it will be realized and that anonymity will be strictly respected. Anonymization was achieved before transmission to CRVOI (Reunion Island) and the only information made available to this laboratory was the age, gender and the island of origin for each sample.

Study design and sample collection

This is a cross sectional study involving 400 individuals living on Ngazidja (n = 196), Ndzouani (88) and Mwali (116), the three main islands of the Union of Comoros. Anonymous participants were 325 individuals suffering diverse medical conditions, consulting either private laboratories of the three islands or the Surveillance Laboratory for Malaria (PNLP) located on Ngazidja. Patients consulting private laboratories had heterogeneous conditions and were mainly suffering from chronic diseases (diabetes, hypertension etc.) or occasionally fever, while PNLP patients were mostly febrile. Seventy-five healthy individuals accompanying patients also accepted to participate to the study and represented 22, 13 and 18% of all individuals enrolled on Ngazidja, Mwali and Ndzouani, respectively. The study was conducted from August 1–October 8, 2011 and was based on excess sera remaining after laboratory tests motivating the visit to the medical center were performed. The eligibility criteria were acceptance to participate to the study and age over 15 years. Sera were stored at -20°C until testing.

The number of required sera from each island was calculated taking into account expected seropositivity rates based on literature [4,7,11] in order to be able to show a difference of 20% between islands for CHIKV and DENV, and of 6% for RVFV. Hence, the minimum sample size was estimated to 100 per island for CHIKV and DENV and to 190 per island for RVFV (S1 Table). The numbers of sera actually collected and included in the present study were 196, 116 and 88 originating from Ngazidja, Mwali and Ndzouani, respectively.

ELISA-based detection of antibodies to arboviruses

IgM antibodies to arboviruses. In-house immunoglobulin (Ig) M-capture enzyme immunoassays (MAC-ELISA) were used to detect serum IgM antibodies to CHIKV, DENV, WNV and RVFV. Briefly, IgM antibodies were captured with rabbit anti-human IgM antibodies (Interchim, Montluçon, France). CHIKV, DENV, WNV and RVFV antigens were prepared on Vero cells in a BSL-3 environment, precipitated and inactivated by betapropiolactone (Sigma-Aldrich, St Quentin Fallavier, France). Specific antigen binding was demonstrated by using an ascitic fluid from mouse hyperimmune to CHIKV, DENV, WNV, and RVFV, respectively and a goat anti-mouse peroxidase-labeled conjugate (Interchim). Serum samples were considered positive if the optical density at 450 nm was >3-fold the mean of negative sera with a Sunrise spectrophotometer (Tecan, Lyon, France) [12].

IgG antibodies to arboviruses. The screening of antibodies (IgG) against various pathogens was performed using both in house (Yellow fever virus (YFV), CHIKV and RVFV) and commercial (DENV, Tick-borne encephalitis virus (TBEV) and West Nile virus (WNV)) enzyme linked immune-absorbent assay (ELISA) protocols. Yellow fever virus, CHIKV and RVFV assays were described elsewhere [13] and relied on the use of antigen derived from whole-virion particles in non-inactivated cell culture supernatants. They were operated in a BSL-3 environment. Commercial kits were used according to the manufacturers' recommendations for detection of antibodies to DENV (PanBio[®], Brisbane, Australia), WNV and TBEV (EuroImmun[®], Lübeck, Germany). Positive and negative control sera were provided by the French National Reference Centre for Arboviroses or by the manufacturers. For each serological assay, a minimum of 3 positive controls was included, alongside 3 negative and 3 blank samples (normal saline) used as controls. For consistence, all samples were tested in duplicates using common serum controls (negative and positive) for all plates in a specific pathogen assay. The values of all plates for a given test were subsequently normalized according to values

of negative and positive controls. In addition, results from a panel of 176 true negative samples (based on sero-neutralisation methods) were used for determining cut-off values of both inhouse and commercial kits [14]. Sera with normalized absorbance values above the cut-off value (defined as [mean of normalized true negatives + two standard deviations]) were considered to be positive (Absorbance Ratio (AR)>1.1). As members of the flaviviridae (*i.e.* DENV, WNV, TBEV and YFV) are to a large part sero crossreactive, a more stringent criteria was applied to the data pertaining to this viral family in order to identify among the four viral species the most probable etiologic virus. The latter was defined as the virus giving the AR>1.1 plus at least 0.5 over the AR of any the 3 other flaviviruses.

DENV sero-neutralisation tests

DENV serotyping was performed using an ELISA-format micro neutralizations test as previously described [15] In this assay, viral neutralization is not measured by the reduction of the number of lysis foci, but rather by the reduction of viral proteins produced in the plate, as detected with a spectrophotometer through an indirect immunoperoxydase antibody test. This assay has been shown to provide monotypic responses similar to the standard serotype-specific PRNT assay in serum post primary infection [16]. Briefly, viruses were grown in 96-well plates in Vero E6 monolayers in the presence of two-fold serial dilutions of serum ranging from 1/40 to 1/2560. Seven days post-infection, supernatants were removed; cells were fixed with 4% para-formaldehyde and permeabilized with 0.5% Triton X-100. Viral proteins were detected spectrophotometrically using a commercially available pan-flavivirus anti-NS5 monoclonal antibody from hybridoma H86.13 B4A supernatant, HRP-conjugated anti-mouse secondary antibodies and TMB as peroxidase substrate. For each serum and virus, the neutralization titer was the reciprocal of the highest serum dilution that inhibited virus protein production, *i.e.* leading to a mean OD below the positive cut-off value as determined with the corresponding control test DENV. The DENV serotype that is neutralized at the highest serum titer is referred as the dominant serotype. Negative controls were performed using four true negative samples (courtesy of the French National Reference Centre for Arboviruses). The four DENV serotypes used in the assay were H/IMTSSA/98/060, H/IMTSSA-MART/98-703, H87, and Dak HD 34 460, respectively [16-19] Sera with neutralizing titer \geq 40 were considered positive.

Statistical analysis

Seropositivity rates were first calculated for each arbovirus on the whole sample, then according to island, age, and clinical condition ("healthy" versus "unhealthy" individuals) and compared together using chi-square or Fisher's exact test.

Results

From August 1st to October 8th of 2011, 400 excess sera were collected in private laboratories and PNLP, from 75 healthy individuals and 325 individuals suffering various clinical conditions (Sex Ratio M/F = 0.61) who sought laboratory testing for diverse medical reasons, including blood smear analysis for malaria diagnosis in febrile patients. Characteristics of the population sample according to age class, gender and living place, are provided in Table 1. A total of 196 (49.0%), 116 (29.0%), and 88 (22.0%) subjects originated from Ngazidja, Mwali and Ndzouani, respectively. Sixty seven (16.7%), 133 (33.3%), 131 (32.7%) and 69 (17.3%) sera were from children (<15years old (yo)), young adults (15–30 yo); adults (31–50 yo) and older adults (>50 yo), respectively.

IgG Seropositivity rates are reported on Table 2 for each arbovirus in the whole sample and according to the living Island. Data show that most individuals (75%) have been previously

	<15yo	15-30yo	31-50yo	>50yo	All age classes
Ngazidja	41	52	59	44	196
Females, Males	23, 18	32, 20	36, 23	25, 19	116, 80
Mwali	4	42	51	19	116
Females, Males	3, 1	34, 8	30, 21	8, 11	75, 41
Ndzouani	22	39	21	6	88
Females, Males	9, 13	31, 8	15, 6	3, 3	58, 30
Total	62	133	131	74	400

Table 1. Distribution of the study sample according to gender, age class and island.

doi:10.1371/journal.pntd.0004840.t001

infected by DENV while seropositivity rates to CHIKV, RVFV and WNV were much lower (12.0%, 10.7% and 7.2%, respectively). Finally, very few individuals were seropositive for TBEV and YFV (0.7% and 0.5%, respectively). When the data generated from individuals suffering clinical conditions (n = 325) and from healthy individuals (N = 75) were analyzed separately (Table 2), the prevalence rates were not statistically different between the two subgroups for DENV, CHIKV, YFV and WNV but were found significantly higher for RVFV and TBEV in the subgroup of healthy individuals.

In addition to IgG serology that was carried out on the whole sample, there was enough residual serum to screen for the presence of IgM antibodies against DENV, CHIKV, WNV and RVFV viruses for 364 samples. Very few sera were found seropositive for IgM to DENV, WNV and CHIKV (3, 2 and 0 sera respectively) (Table 3). In contrast, higher rates of seropositivity for IgM to RVFV were recorded in the three islands with the highest rates on Mwali (10.25%), and Ndzouani (8.04%) compared to Ngazidja (2.5%) (p<0.05).

IgG seropositivity rates to DENV

A total of 300 sera tested positive through ELISA for IgG antibodies to DENV. This consisted of 174, 77 and 49 sera from Ngazidja, Mwali and Ndzouani, corresponding to 88.8%, 66.4%

Arbovirus	Ngazidja n = 196	Mwali n = 116	Ndzouani n = 88	Total n = 400	р	Healthy group n = 75	Unhealthy group n = 325	р
	n (%)	n (%)	n (%)	n (%)		n (%)	n (%)	
CHIKV	37 (18.9)	9 (7.8)	2 (2.3)	48 (12.0)	<0.0001	12 (16.0)	36 (11.0)	0.24
RVFV	31 (15.8)	6 (5.2)	6 (6.8)	43 (10.7)	0.005	17 (22.7)	26 (8.0)	<0.001
DENV	174 (88.8)	77 (66.4)	49 (55.7)	300 (75.0)	<0.0001	54 72.0)	235 (72.3)	0.96
TBEV	3 (1.5)	0 (0.0)	0 (0.0)	3 (0.7)	0.20	3 (4.0)	0 (0.0)	<0.01
YFV	2 (1.0)	0 (0.0)	0 (0.0)	2 (0.5)	0.35	0 (0.0)	2 (0.6)	0.96
WNV	6 (3.1)	21 (18.1)	2 (2.3)	29 (7.3)	<0.0001	6 (8.0)	14 (4.3)	0.19

Table 2. IgG antibodies to each arbovirus according to living Island and to the status of serum donors (n = 400). The positivity criteria is a Absorbance Ratio to the cut-off value >1.1.

doi:10.1371/journal.pntd.0004840.t002

Table 3. IgM antibodies to DENV, WNV, CHIKV and RVFV in 364 sera collected from Grande Comore, Moheli and Anjouan.

Arbovirus	Ngazidja n positive /n tested (prevalence in the sample)	Mwali n positive /n tested (prevalence in the sample)	Ndzouani n positive /n tested (prevalence in the sample)	Total (prevalence in the sample)		
DENV	2/ 160 (1,25%)	1/117 (0.85%)	0/87 (0%)	3/364 (0.82%)		
WNV	0/160 (0%)	2/117 (1.70%)	0/87 (0%)	2/364 (0.55%)		
CHIKV	0/160 (0%)	0/117 (0%)	0/87 (0%)	0/364 (0%)		
RVFV	4/160 (2.5%)	12/117 (10.25%)	7/87 (8.04%)	23/ 364 (6.31%)		

doi:10.1371/journal.pntd.0004840.t003

Table 4. Number of sera with neutralizing antibodies to DENV serotypes 1–4 according to (i) the neutralized serotype(s), (ii) the dominant serotype, (iii) the Island and (iv) the age class. Presented data concern a subset of 90 randomly selected DENV ELISA positive samples. (* Numbers may not add up as one individual serum may react with more than one serotype).

DENV Serotype	Serotype	Serotype	Serotype	Serotype
	1	2	3	4
i) Number* of sera with neutralizing antibodies to the indicated DENV serotype:	57 (63.3%)	47 (52.2%)	68 (75.6%)	44 (48.9%)
ii) Number* of sera with neutralizing antibodies to the indicated serotype considered as dominant serotype:	35 (38.9%)	21 (23.3%)	31 (34.4%)	14 (15.6%)
iii) Island:				
Grande Comore (Ngazidja)	25	15	25	7
Moheli (Mwali)	8	4	4	6
Anjouan (Ndzouani)	2	2	2	1
iv) Age class:				
<15yo	8	6	16	1
15-30уо	12	4	8	4
30-50уо	15	11	7	9

doi:10.1371/journal.pntd.0004840.t004

and 55.7% seropositivity rates, respectively. The difference in seropositivity according to island was highly significant (p<0.0001).

As several dengue epidemics hit the Comoros archipelago during the last 70 years, we typed sera against each of the 4 DENV serotypes using the ELISA-format micro neutralization test derived from the plaque reduction neutralization test (PRNT). A subgroup of 90 sera testing ELISA positive for DENV was randomly selected from persons aged less than 50. Sera from individuals over age 50 were not considered for this test since interpretation of the neutralization profiles is more complicated in older age classes which are more likely to have experienced sequential infections by different DENV serotypes, thus inducing stronger heterotypic DENV neutralization responses. Four samples testing ELISA negative were used as controls. Twelve sera tested negative by the ELISA-format micro neutralization test despite being positive through ELISA to DENV antigen, and were not further considered. Seventy-eight sera (54, 18 and 6 originating from Ngazidja, Ndzouani and Mwali, respectively) had individual titers over 40 to one or several DENV serotype(s) and were considered positive to the reactive serotype (Table 4). The mean titer of reactivity was 1093, 835, 566 and 718, for serotypes 1 to 4, respectively. Noteworthy, only few sera (14 out of 78), reacted with one single serotype (Table 5) If one considers only the dominant serotypes (*i.e.* the serotype identified at the highest titer by each serum sample), the most frequently recognized serotypes were DENV serotypes 1 and 3,

Table 5. Neutralizing antibodies for DENV serotypes 1–4. Reactivity profile according to the number of seroneutralized serotypes detected in 78 out of 90 randomly selected DENV ELISA positive samples.

Island of origin and age class	Grande Comore (Ngazidja)	Moheli (Mwali)	Anjouan (Ndzouani)	Total tested	<15yo	15-30yo	30-50yo
Number of sera with neutralizing antibodies to DENV serotypes 1–4	54	18	6	78	23	22	33
Number of sera with neutralizing antibodies to DENV according to the number of reactive serotypes							
1 serotype	5 (9.2%)	6 (33.3%)	3 (50.0%)	14	4 (17.4%)	6 (27.3%)	4 (12.1%)
2 serotypes	17 (31.5%)	5 (27.8%)	0 (0.0%)	22	10 (43.5%)	7 (31.8%)	5 (15.1%)
3 serotypes	12 (22.2%)	3 (16.7%)	1 (16.6%)	16	6 (26.1%)	2 (9.1%)	8 (22.2%)
4 serotypes	20 (37.0%)	4 (22.2%)	2 (33.2%)	26	3 (13.0%)	7 (31.8%)	16 (48.5%)

doi:10.1371/journal.pntd.0004840.t005

particularly on Ngazidja (Table 4). Noteworthy, serotype 3 was mostly recognized in sera from young persons, and serotype 1 from older individuals.

IgG seropositivity rates to CHIKV, RVFV and WNV

A total of 48 sera were positive for IgG antibodies to CHIKV. A significant difference was observed according to the sampled island (p = 0,0001), with a seropositivity rate higher on Ngazidja (18.9%) than on Mwali (7.8%) and Ndzouani (2.3%). No difference in seroprevalence was observed according to age.

Forty-three sera (10.7%) were positive for IgG antibodies to RVFV with seropositivity rates being again statistically different according to the sampled island (p<0.001). Sera from Ngazidja, Mwali and Ndzouani displayed IgG seropositivity rates to RVFV of 15.8%, 5.2% and 6.8% (see Table 2), respectively. No significant difference in seroprevalence was observed according to age, although it tended to be higher in adults >50 yo (18.9%) than in the three other age groups (10.4%, 9.8% and 7.6%, respectively). Finally, 29 sera were positive for IgG antibodies to WNV by ELISA, with most of them (21/29) originating from Mwali (Table 2). Using the most stringent criteria to discriminate between infections by members of flaviviridae family (See Material and Methods section), 21 among these 29 sera were unambiguously positive for WNV (S1 Table), 7 sera yielded ambiguous results between WNV and DENV and one serum between WNV and TBEV. Interestingly, most (15/21) sera unambiguously testing positive for WNV still originated from Mwali. This was also the case for 5 out of the seven sera with ambiguous positivity to both DENV and WNV (S1 Table). These results support the selectivity of WNV infection on Mwali.

IgG antibodies to tick-borne encephalitis and yellow fever virus

Three sera were positive for TBEV and two for YFV and they were all originated from Ngazidja. Altogether, among the 400 sera composing the serobank, 338 (84.5%) reacted with at least one of the investigated arboviruses. The seropositivity rate to at least one arbovirus showed a strong association with age (p<0.0001) since it was 68.7%, 81.2%, 90.1%% and 95.7% in the four successive age groups (<15yo; 15-30yo; 31-50yo; >50yo), respectively.

Discussion

IgG antibodies to arboviruses are long lasting signatures of the arboviral outbreaks that have been hitting inhabitants of the Comoros Islands for decades. Almost 85% of sera investigated in the present study reacted with at least one arbovirus and the seropositivity rate increased with age, stressing the intense and long lasting exposure of the Comorian population to arboviral risk. As demonstrated by the almost negative results for IgM antibodies, this study was conducted during an inter-epidemic phase for DENV, CHIKV and WNV infections. However for RVFV, we had evidence of recent or current circulation of RVFV on the whole country though mostly on the two southern islands of the Union of Comoros. The two subgroups composing the whole sample used in the study (*i.e.* "not healthy" individuals consulting private or public laboratories for various medical reasons n = 325; and "healthy" individuals (n = 75) had prevalence rates not statistically different for DENV, CHIKV, YFV and WNV but significantly higher for RVFV and TBEV in the control group. At the very least, this result argues that our data were not biased towards more detection of arboviral infections in the "not healthy" individuals (especially the febrile ones) compared to "healthy" individuals. Our data also indicate that the most significant risks pertain to DENV, CHIKV, RVFV and WNV infections, with DENV exposure clearly appearing as the most prominent. A strict criterion was used to define positive sera based on a panel of true negative samples that were confirmed by sero neutralization.

Noteworthy, the use of the more stringent criteria to identify the likely infecting virus among the flaviviruses (i.e. DENV, WNV, TBEV and YFV), yielded essentially similar results (see S1 Table). Titration of neutralizing antibodies to the four DENV serotypes indicates that most sera react with more than one serotype, an additional confirmation of the iterative epidemics bouts hitting these islands. Sera were most frequently reactive with DENV1 and DENV3 and less frequently with DENV2 and DENV4. These results are in keeping with the history of dengue epidemics on the archipelago during the last seventy years: DENV1 epidemics was recorded in 1948 and 1993, DENV2 in 1984 [20] and DENV3 in 2010 [21] To our knowledge, serotype 4 has never been reported as causing an epidemic in the Indian Ocean region though some uncertainty remains with regard to serotypes that have been circulating during the 1948 and 1984 epidemics in Comoros [22]. Yet, neutralizing antibodies to DENV4 were detected in the present series in over half of titrated sera (44/78). As for sera neutralizing multiple serotypes, this DENV4 positivity may reflect a mere cross reactivity. However, 14 sera expressed the highest neutralizing titer to this serotype 4 and these sera were mostly (9 out of 14) from adults aged over 30. It is therefore probable that DENV4 has been circulating at low rate, presumably several decades ago.

Considering all DENV serotypes, our data bring in further evidence that human populations living on Ngazidja have been more exposed to dengue epidemics than those living on Mwali or Ndzouani. In the present study, IgG antibodies to DENV were detected in 89%, 66% and 56% of sera from individuals living in Ngazidja, Mwali and Ndzouani, respectively, the difference in seropositivity rates being highly significant. In addition, most sera reacting with only one DENV serotype were from Mwali and Ndzouani (9/14) whereas most sera from Ngazidja reacted with multiple serotypes and over one third of them with all four DENV serotypes. These data are in keeping with previous reports showing that during the epidemics of 1993, rates of IgG positivity were 83% on Ngazidja, 40% on Mwali and 22.5% on Ndzouani [4].

It is likely that compared to Ngazidja, the relative preservation from arboviral risk of the southern islands of the archipelago also holds true for Mayotte. In that island, a cross-sectional serologic survey conducted in 2006 during an inter-epidemic period for dengue, showed a rate of DENV-specific IgG antibodies of 22.7% with an age-specific prevalence peaking at 38.8% in the 25–54 years age group while the 2–14 age group was almost completely preserved [20]. Interestingly, Comorian native inhabitants aged 15 years or over, who had immigrated to Mayotte, were around ten-times more affected than individuals of the same age born in Mayotte [20].

The seroneutralisation analysis carried out in the present study on a subset of dengue ELISA positive samples indicates that almost all of these sera actually contained antibodies neutralizing DENV, with two thirds neutralizing 3 or 4 serotypes. Hence, a large fraction of the population (>40%) could be considered as currently protected against dengue. This could explain the long interval (several decades) between two epidemic bouts, allowing immunologically naive individuals (mainly young persons) to accumulate into the pool of susceptible until the next dengue outbreak flares up. These periodic bursts do not preclude a low level of DENV transmission in the Union of Comoros during inter-epidemic periods; however proof of such low-rate transmission is missing as most dengue like syndromes are not investigated in this resource limited country.

With regard to Chikungunya virus, our results indicate a similar trend toward a more severe exposure of Ngazidja to this arbovirus compared to the two other islands. Our study has detected IgG seropositivity rates to CHIKV of 18.9% on sera from Ngazidja and 5.8% and 2.3% on sera from Mwali and Ndzouani, respectively. Surprisingly, the seropositivity rate on Ngazidja appeared rather low compared to the attack rate of 63% (60% IgM antibodies, 27% IgG) reported on this island during the large epidemic of Chikungunya in 2005 [7] In fact, the

IgG seropositivity rates reported by Sergon and co-workers [7] and the one detected in the present study are relatively close and the contrast is mainly created by the very high level of IgM reported during this epidemic. This apparent discrepancy suggests either that IgM positivity to CHIKV was overestimated by Sargon *et al.* or that a large fraction of IgM positive individuals actually failed to efficiently switch into high titres and long lasting IgG antibodies to CHIKV. Few data have been reported so far in the literature on the long-term persistence of antibodies to CHIKV following infection. However, almost 10% of seropositive sera from individuals infected on Reunion Island during the 2006 epidemics reverted to ELISA negative within 3 years after the epidemics (A. Michault, personal communication). In Thailand, nine-teen years after a CHIKV outbreak occurred, over one third of 111 former patients had neutralizing antibodies reacting against CHIKV antigen [23]

In addition to showing maximal exposure of Ngazidja inhabitants to DENV and CHIKV, our study brings in the first serological evidence for RVFV circulation amongst humans on the islands of the Union of Comoros. This was demonstrated not only by high rates of IgG seropositivity to the virus but also by the detection of IgM seropositivity indicating recent or current circulation of the virus mainly through pauci- or a-symptomatic infection. Rift Valley fever is a vector borne zoonotic disease present in Africa, the Middle East, and Madagascar [24-28]. Our data reveal higher IgG seropositivity rates in Ngazidja but higher IgM seropositivty rates in Mwali and Ndzouani. These results confirm the only suspicion going back to 2007 when an autochthonous clinical case of RVFV was detected in a child from Ngazidja admitted at Mayotte's hospital with a 2-month history of severe encephalitis [28]. They are also coherent with the demonstration of an active circulation of the virus in Comorian livestock [29,30]. With regard to Mayotte, an investigation carried out on 2007–2008 on denguelike syndromes, testing negative for Plasmodium spp., CHIKV, and DENV, revealed 4.5% positive cases, suggestive of recent RVFV infection, *i.e.* presence of viral RNA or IgM [31]. A serosurvey study (2011) conducted also on Mayotte on 1420 individuals showed an overall seroprevalence of 3.5% for RVFV antibodies in the general population aged over 5 years. Altogether these results indicate active circulation, though mostly clinically insidious, of RVFV in humans in the four islands of the Comoros archipelago.

Finally, our study is the first reporting on WNV infection in the Comoros archipelago. Several studies have shown active circulation of WNV on Madagascar [32–34] and Seychelles [35], and the virus has been isolated at several occasions from birds in Madagascar [33]. Interestingly the island of Mwali clustered 75% of sera testing positive to this flavivirus in our study, indicating that among the Comorian population, inhabitants of Mwali bear the maximal risk of WNV infection. The use of stringent criteria to discriminate true WNV infection, as well as the observation that most seropositive sera clustered in the smallest island of the Union of Comoros are strong arguments against a mere cross reactivity with other flaviviruses. However, the formal proof would require testing for neutralizing antibodies.

Thus, our study highlights the distinct exposure of the different Comorian islands to arboviral risk and one may address the question on which factors could account for such peculiarity. Geographic, ecologic, entomologic and anthropogenic factors are most likely the determinants of such heterogeneity for which we propose several non-exclusive hypotheses.

Geographical and anthropogenic factors

Ngazidja, the closest to the African continent among the archipelago islands, is directly exposed to the East African coast, a hot spot for the emergence of zoonotic and vector borne diseases [36] Moreover, Ngazidja is the entry point for active trade with continental African countries conveying pathogens and vectors. For example, the bilateral trade bill established in

2000 with Tanzania had allowed importation of cattle to Ngazidja and was soon followed by the emergence of an epidemic of theileriosis on this island likely due to the introduction of the infected tick vector [37,38]. Similarly, the higher IgG seropositivity rates to CHIKV, DENV and RVFV measured on Ngazidja in the present study may simply reflect this geographical proximity. However, the higher IgM positivity rates to RVFV in sera from Mwali is in keeping with the fact that livestock was mostly infected on Mwali [29,30].

Geological and entomological factors

Comorian islands are volcanic islands spanning along a South/North line which ages range from 7.7–15 yo in the eldest Mayotte to 0.1–0.5 yo in the youngest Ngazidja [3]. These stretched ages impact their geologic structure and shape the ecologic conditions that prevail on these islands. For instance, Ngazidja has a very shallow soil layer, which cannot hold water, while the other islands, geologically older, support more advanced soil lateritisation that allows better persistence of surface water. The absence of perennial river on Ngazidja due to the highly porous soil imposes rainfall water to be stored in catchment tanks, creating anthropogenic conditions that are conducive to the proliferation of mosquito larvae. Subtle differences may exist between mosquito populations in the different islands. For instance, several mitochondrial haplotypes displaying some level of geographical structuration have been characterized among natural populations of *Ae. aegypti* sampled throughout other islands of the South West Indian Ocean and these haplotypes may be associated with distinct morphological and ecological traits [39]. Such information, as well as studies on the vector competence of these distinct lineages towards CHIKV and DENV are not available on the Union of the Comoros and should help understanding the human serological figures highlighted here.

The present study shows that Mwali is clearly more exposed to WNV infection than the two sister islands. It is unlikely that this peculiarity relies on significantly different exposure to the bites of *Culex* spp., the usual vectors of WNV. It rather suggests, if one considers the major role played by birds in the transmission of this virus, some specificities of Mwali with regard to local birds populations. In fact, Mwali is known to host the major seabird reproduction colonies of the whole Comorian islands [40]. Further investigation of migratory routes together with an analysis of the serological status of both migratory and local bird populations will help clarifying this issue.

The main limitation of our investigation is in the study design, which for feasibility reasons was based on the use of excess sera collected from laboratory consultants. Although enrolled participants reside in either of the three main islands, they were not representative of the whole Comorian population. However our results are in keeping with the scarce studies conducted so far in this country indicating an intense exposure of all islands of the Union of Comoros to arboviral infections with salient disparities among islands with regard to some arboviruses. They stress the importance of Ngazidja as the potential entry point for arboviral infections to the whole south Western Indian Ocean region and raise interesting working hypothesis to account for the local specificities of arboviral epidemiology in this tropical island ecosystem. Controlled studies are needed to identify risk factors to various arbovirus infections to which the population of the different islands are exposed.

Supporting Information

S1 Table. Identification of the dominant seroreactive flavivirus in the whole sample according to the living island, Union of Comoros (n = 400). The dominant seroreactive flavivirus is the one with an AR>1.1 plus 0.5 higher than the data of other flaviviruses. (DOC)

S1 Checklist. STROBE Statement—Checklist of items that should be included in reports of *cross-sectional studies*.

(DOC)

Acknowledgments

The Ministère de la Santé, de la Solidarité et de la Promotion du Genre of the Union of the Comoros is acknowledged for his help in setting up the study.

Author Contributions

Conceived and designed the experiments: KD XdL PT.

Performed the experiments: NS AY RS ILG MM.

Analyzed the data: SL.

Contributed reagents/materials/analysis tools: NS AY RS ILG MM XdL.

Wrote the paper: PT KD.

References

- Afrane YA, Little TJ, Lawson BW, Githeko AK, Yan G. Deforestation and vectorial capacity of *Anopheles gambiae* giles mosquitoes in malaria transmission, Kenya. Emerg Infect Dis. 2008; 14: 1533–1538. doi: 10.3201/eid1410.070781 PMID: 18826815
- 2. Paaijmans KP, Blanford S, Chan BHK, Thomas MB. Warmer temperatures reduce the vectorial capacity of malaria mosquitoes. Biol Lett. 2012; 8: 465–468. doi: 10.1098/rsbl.2011.1075 PMID: 22188673
- Tortosa P, Pascalis H, Guernier V, Cardinale E, Le Corre M, Goodman SM, et al. Deciphering arboviral emergence within insular ecosystems. Infect Genet Evol. 2012; 12: 1333–1339. doi: 10.1016/j.meegid. 2012.03.024 PMID: 22504353
- Boisier P, Morvan J, Laventure S, Charrier N, Martin E, Ouledi A, et al. Epidemie de Dengue 1 sur l'île de la Grande Comore. République fédérale islamique des Comores. Mars-Mai 1993. Ann Soc Belge Med Trop. 1994; 74: 217–229.
- Lernout T, Giry C, Binder P, Zumbo B, Durquety E, Lajoinie G, et al. Emergence of Dengue virus serotype 3 on Mayotte Island, Indian Ocean. East Afr J Public Health. 2011; 8: 155–156. PMID: 22066303
- Sergon K, Njuguna C, Kalani R, Ofula V, Onyango C, Konongoi LS, et al. Seroprevalence of Chikungunya virus (CHIKV) infection on Lamu Island, Kenya, October 2004. Am J Trop Med Hyg. 2008; 78: 333– 337. PMID: 18256441
- Sergon K, Yahaya A, Brown J, Bedja S, Mlindasse M, Agata N, et al. Seroprevalence of Chikungunya virus infection on Grande Comore Island, Union of the Comoros, 2005. Am J Trop Med Hyg. 2007; 76: 1189–1193. PMID: 17556634
- 8. Mlindasse M. International symposium: Chikungunya and other emerging arboviroses in tropical areas. Chikungunya in Comoros. St Pierre, La Réunion; 2007.
- Sissoko D, Malvy D, Giry C, Delmas G, Paquet C, Gabrie P, et al. Outbreak of Chikungunya fever in Mayotte, Comoros archipelago, 2005–2006. Trans R Soc Trop Med Hyg. 2008; 102: 780–786. doi: 10. 1016/j.trstmh.2008.02.018 PMID: 18400240
- 10. Schuffenecker I, Iteman I, Michault A, Murri S, Frangeul L, Vaney M-C, et al. Genome Microevolution of Chikungunya Viruses Causing the Indian Ocean Outbreak. PLoS Med. 2006; 3.
- Morvan J, Saluzzo JF, Fontenille D, Rollin PE, Coulanges P. Rift Valley fever on the east coast of Madagascar. Res Virol. 1991; 142: 475–482. PMID: 1687082
- Peyrefitte CN, Couissinier-Paris P, Mercier-Perennec V, Bessaud M, Martial J, Kenane N, et al. Genetic characterization of newly reintroduced Dengue virus type 3 in Martinique (French West Indies). J Clin Microbiol. 2003; 41: 5195–5198. doi: 10.1128/JCM.41.11.5195-5198.2003 PMID: 14605161
- **13.** andayi F, Charrel R, Kieffer A, Richet H, Pastorino B, Leparc-Goffart I, et al. A sero-epidemiological study of arboviral fevers in Djibouti, horn of Africa. PLoS Negl Trop Dis. 2014; In press.
- 14. L'Azou M, Jean-Marie J, Bessaud M, Cabié A, Césaire R, de Lamballerie X, et al. Dengue Seroprevalence in the French West Indies: A Prospective Study in Adult Blood Donors. Am J Trop Med Hyg. 2015;

- Vorndam V, Beltran M. Enzyme-linked immunosorbent assay-format microneutralization test for dengue viruses. Am J Trop Med Hyg. 2002; 66: 208–212. PMID: 12135295
- Tolou HJ, Couissinier-Paris P, Durand JP, Mercier V, de Pina JJ, de Micco P, et al. Evidence for recombination in natural populations of Dengue virus type 1 based on the analysis of complete genome sequences. J Gen Virol. 2001; 82: 1283–1290. doi: 10.1099/0022-1317-82-6-1283 PMID: 11369871
- Tolou H, Couissinier-Paris P, Mercier V, Pisano M-R, de Lamballerie X, de Micco P, et al. Complete Genomic Sequence of a Dengue Type 2 Virus from the French West Indies. Biochem Biophys Res Commun. 2000; 277: 89–92. doi: 10.1006/bbrc.2000.3640 PMID: 11027645
- Osatomi K, Sumiyoshi H. Complete nucleotide sequence of Dengue type 3 virus genome RNA. Virology. 1990; 176: 643–647. PMID: 2345967
- Moureau G, Temmam S, Gonzalez JP, Charrel RN, Grard G, de Lamballerie X. A real-time RT-PCR method for the universal detection and identification of flaviviruses. Vector Borne Zoonotic Dis Larchmt N. 2007; 7: 467–477.
- Sissoko D, Ezzedine K, Giry C, Moendandzé A, Lernout T, D'Ortenzio E, et al. Seroepidemiology of Dengue virus in Mayotte, Indian Ocean, 2006. PloS One. 2010; 5: e14141. doi: 10.1371/journal.pone. 0014141 PMID: 21152441
- 21. Mlindasse M. Investigation d'une épidémie de engue type 3 aux Comores, mars 2010. Antananarivo, Madagascar; 2012. http://www.reseausega-coi.org/system/files/06-Mlindasse-DENV3_0.pdf
- 22. Zeller HG. Dengue, arboviruses and migrations in the Indian Ocean. Bull Soc Pathol Exot 1990. 1998; 91: 56–60.
- Nitatpattana N, Kanjanopas K, Yoksan S, Satimai W, Vongba N, Langdatsuwan S, et al. Long-term persistence of Chikungunya virus neutralizing antibodies in human populations of North Eastern Thailand. Virol J. 2014; 11: 183. doi: 10.1186/1743-422X-11-183 PMID: 25330992
- Turell MJ, Linthicum KJ, Patrican LA, Davies FG, Kairo A, Bailey CL. Vector competence of selected African mosquito (Diptera: Culicidae) species for Rift Valley fever virus. J Med Entomol. 2008; 45: 102– 108. PMID: 18283949
- Madani TA, Al-Mazrou YY, Al-Jeffri MH, Mishkhas AA, Al-Rabeah AM, Turkistani AM, et al. Rift Valley Fever Epidemic in Saudi Arabia: Epidemiological, Clinical, and Laboratory Characteristics. Clin Infect Dis. 2003; 37: 1084–1092. doi: 10.1086/378747 PMID: 14523773
- World Health Organization. Outbreaks of Rift Valley fever in Kenya, Somalia and United Republic of Tanzania, December 2006-April 2007. Wkly Epidemiol Rec. 2007; 82: 169–178. PMID: 17508438
- 27. Andriamandimby SF, Randrianarivo-Solofoniaina AE, Jeanmaire EM, Ravololomanana L, Razafimanantsoa LT, Rakotojoelinandrasana T, et al. Rift Valley fever during rainy seasons, Madagascar, 2008 and 2009. Emerg Infect Dis. 2010; 16: 963–970. doi: 10.3201/eid1606.091266 PMID: 20507747
- Sissoko D, Giry C, Gabrié P, Tarantola A, Pettinelli F, Collet L, et al. Emergence of Rift Valley fever in Mayotte, 2007–2008. BEH. 2009: 33–36.
- Roger M, Girard S, Faharoudine A, Halifa M, Bouloy M, Cêtre-Sossah C, et al. Rift Valley Fever in Ruminants, Republic of Comoros, 2009. Emerg Infect Dis. 2011; 17: 1319–1320. doi: <u>10.3201/eid1707</u>. 102031 PMID: 21762604
- Roger M, Beral M, Licciardi S, Soule M, Faharoudine A, Foray C, et al. Evidence for Circulation of the Rift Valley Fever Virus among Livestock in the Union of Comoros. PLoS Negl Trop Dis. 2014; 8.
- Sissoko D, Giry C, Gabrie P, Tarantola A, Pettinelli F, Collet L, et al. Rift Valley fever, Mayotte, 2007–2008. Emerg Infect Dis. 2009; 15: 568–570. doi: 10.3201/eid1504.081045 PMID: 19331733
- 32. Mathiot C, Clerc Y, Rodhain F, Digoutte J, Coulanges P. Le virus West-Nile et Madagascar. Arch Inst Pasteur Madagascar. 1984; 51: 113–124.
- Lonchampt C, Migliani R, Ratsitorahina M, Rabarijaona LP, Ramarokoto CE, Rakoto Andrianarivelo M, et al. Persistence of an endemic circulation of the West Nile virus in Madagascar. Arch Inst Pasteur Madagascar. 2003; 69: 33–36. PMID: 15678813
- 34. Larrieu S, Cardinale E, Ocquidant P, Roger M, Lepec R, Delatte H, et al. A fatal neuroinvasive West Nile virus infection in a traveler returning from Madagascar: clinical, epidemiological and veterinary investigations. Am J Trop Med Hyg. 2013; 89: 211–213. doi: 10.4269/ajtmh.12-0003 PMID: 23751400
- **35.** Yersin C, Bovet P, Herminie P, Zeller H. Public health importance of mosquito-borne diseases in the Seychelles (Indian ocean). Infectious Dis Rev. 1999; 1: 189–199.
- Jones KE, Patel NG, Levy MA, Storeygard A, Balk D, Gittleman JL, et al. Global trends in emerging infectious diseases. Nature. 2008; 451: 990–993. doi: 10.1038/nature06536 PMID: 18288193
- De Deken R, Martin V, Saido A, Madder M, Brandt J, Geysen D. An outbreak of East Coast Fever on the Comoros: a consequence of the import of immunised cattle from Tanzania? Vet Parasitol. 2007; 143: 245–253. doi: 10.1016/j.vetpar.2006.08.018 PMID: 16996692

- Yssouf A, Lagadec E, Bakari A, Foray C, Stachurski F, Cardinale E, et al. Colonization of Grande Comore Island by a lineage of *Rhipicephalus appendiculatus* ticks. Parasit Vectors. 2011; 4: 38. doi: 10. 1186/1756-3305-4-38 PMID: 21414194
- 39. Delatte H, Bagny L, Brengue C, Bouetard A, Paupy C, Fontenille D. The invaders: phylogeography of Dengue and Chikungunya viruses Aedes vectors, on the South West islands of the Indian Ocean. Infect Genet Evol J Mol Epidemiol Evol Genet Infect Dis. 2011; 11: 1769–1781.
- **40.** Louette M, Abderamane H, Yahaya I, Meirte D. Atlas des oiseaux nicheurs de la Grande Comore, de Mohéli et d'Anjouan. Tervuren, Belgium: Musée royal de l'Afrique centrale; 2008.